Author's response to reviews

Title: MiR-199a-5p is up-regulated in gastric cancer and functions as an oncogene by targeting klotho

Authors:

Xu-Jun He (hxj0105099@163.com)
Ying-Yu Ma (myy011525@163.com)
Sheng Sheng (yusheng1987@aliyun.com)
Xiao-Ting Jiang (jiangxiaoting2008@aliyun.com)
Yi-Ding Lu (dusili@126.com)
Liang Tao (tao990645@163.com)
Huang-Ping Wang (13957353173@139.com)
Zhi-Ming Hu (hzm6606@hzcnl.com)
Hou-Quan Tao (taohouquan2008@aliyun.com)

Version: 2 Date: 23 September 2013

Author's response to reviews:

Dear Editor,

We have checked and revised the manuscript according to the reviewers’ comments as follows:

Question 1: Requesting ethics statement:
Answer: We have added ethics statement in method section of the manuscript.

Question 2: Please include an Authors' contributions section:
Answer: We included an Authors' contributions section in the manuscript.

Thank you very much for your attention and consideration.

Sincerely yours,

Hou-Quan Tao
Key Laboratory of Gastroenterology of Zhejiang Province
Hangzhou 310014, Zhejiang
China
Phone: 0086-571-85893781
Fax: 0086-571-85131448
E-mail: taohouquan2008@aliyun.com